Author's response to reviews

Title: A coding polymorphism in Matrix Metalloproteinase 9 reduces risk of scarring sequelae of ocular Chlamydia trachomatis infection

Authors:

Angels Natividad (angels.natividad-sancho@lshtm.ac.uk)
Graham Cooke (grahamsooke@hotmail.com)
Martin Holland (mholland@mrc.gm)
Matthew Burton (matthew.burton@lshtm.ac.uk)
Hassan Joof (hmjoof@yahoo.co.uk)
Kirk Rockett (Kirk.rockett@well.ox.ac.uk)
Dominic Kwiatkowski (dominic.kwiatkowski@paediatrics.ox.ac.uk)
David Mabey (david.mabey@lshtm.ac.uk)
Robin Bailey (robin.bailey@lshtm.ac.uk)

Version: 4  Date: 26 April 2006

Author's response to reviews: see over
Dear Sirs,

Thank you for accepting our manuscript MS: 1383133168941857 - A coding polymorphism in Matrix Metalloproteinase 9 reduces risk of scarring sequelae of ocular Chlamydia trachomatis infection- for publication in BMC Medical Genetics

Please find attached the new version of our manuscript, which has been amended according to your formatting requirements:

1- Sentence case has been used for all headings and sub-headings in the manuscript

2- Two manuscript sections have been included in the manuscript ie, Conclusions and Abbreviations.

Please don't hesitate to contact us if the manuscript requires any additional changes

Yours sincerely

Dr Angels Natividad-Sancho
Dr Robin Bailey